Author's response to reviews

Title: Evaluation of wound healing activity of Allamanda cathartica L and Laurus nobilis L extracts on rats

Authors:

Shivananda B Nayak Dr (snayak@fms.uwi.tt)
Poorna K Lingeshwari Dr (poorkishor@yahoo.com)
Sandiford S S (sandiford@yahoo.com)
Vidyasagar Bhogadi (vsbhogadi@yahoo.com)
Andrew Adogwa (adogwa01@yahoo.com)

Version: 2 Date: 17 January 2006

Author’s response to reviews:

The format of the manuscript is changed as per the author instructions of Journal Complementary and Alternative Medicine. The suggestions from the Assistant editor of BMC series is taken in a positive way to improve the quality of written English in our manuscript suitable for peer review.